Author's response to reviews

Title: Development and evaluation of a computerised clinical decision support system for switching drugs at the interface between primary and tertiary care.

Authors:

Markus G Pruszydlo (mp@pruweb.de)
Stefanie U Walk-Fritz (stefanie.walk-fritz@med.uni-heidelberg.de)
Torsten Hoppe-Tichy (torsten.hoppe-tichy@med.uni-heidelberg.de)
Jens Kaltschmidt (jens.kaltschmidt@med.uni-heidelberg.de)
Walter E. Haefeli (walter.emil.haefeli@med.uni-heidelberg.de)

Version: 2 Date: 20 August 2012

Author's response to reviews:

Dear editor,

many thanks for your email! We have added the requested information on ethics approval. As a consequence we have rephrased and expanded the corresponding section (Data collection and analysis, p. 8). we hope that with these changes the manuscript is now ready for peer-review.

Best regards
Walter E. Haefeli, MD